food & nutrition (

research

REVIEW ARTICLE Sodium – a systematic review for Nordic Nutrition Recommendations 2023

Antti Jula*

Department of Clinical Medicine, University of Turku, Turku, Finland

Popular scientific summary

- Sodium plays an important role for the concentration of the fluid outside the cells in the body and for electrical signalling of the nerves, muscles, and heart.
- The major source of sodium in the diet is salt added to processed foods, including bread, meat products, cheese, and ready meals.
- A lack of sodium, hyponatraemia, is seldom caused by dietary restriction.
- Intake of sodium increases blood pressure in a dose-dependent manner.
- High intake of sodium is associated with increased risk of cardiovascular disease and stroke.

Abstract

Blood pressure (BP) rises along with increasing sodium intake from early childhood to late adulthood, and leads to hypertension among most men and women living in Nordic and Baltic countries. Elevated BP is the leading global risk factor for premature deaths and disability-adjusted life-years. A reduction in sodium intake is essential in the prevention of hypertension in individuals, in the lowering of BP levels, in the treatment of hypertensive individuals, and in decreasing risks associated with elevated BP. There is a progressive linear dose-response relationship between sodium intake and BP beginning from a sodium intake of less than 0.8 g/day. Sodium reduction decreases BP linearly by a dose-response manner down to a sodium intake level of less than 2 g/day. Randomised intervention studies with a duration of at least 4 weeks confirm the efficiency and safety of reducing blood sodium intake to a level of less than 2 g/day. Results from prospective cohort studies show that higher sodium intake is positively associated with an increased risk of stroke and cardiovascular events and mortality among the general adult population, and the associations are linear in studies using proper sodium assessment methods. Analyses assessing sodium intake using at least two 24-h urine samples have shown a linear positive relationship between sodium intake and the risk of a cardiovascular event or death. Based on an overall evaluation of the available data, a limitation of the sodium intake to 2.0 g/day is suggested for adults. The optimal sodium intake level would be probably about 1.5 g/day. Sodium intake recommended for children can be extrapolated from the recommended sodium intake for adults. According to national dietary surveys, the average sodium intakes in Nordic countries range in adult men from 3.6 to 4.4 g/day and in adult women from 2.6. to 3.2 g/day, and in Baltic countries in men from 2.6 to 5.1 g/day and in women from 1.8 to 3.6 g/day.

Keywords: sodium; sodium chloride; salt; blood pressure; hypertension; nutrition recommendations

Received: 26 February 2022; Revised: 25 November 2022; Accepted: 25 October 2023; Published: 31 January 2024

odium chloride (NaCl; table salt), the dominant and most important source of dietary sodium, is used as a food ingredient or condiment. Sodium is also found in unprocessed foods, but usually in very low concentrations. One gram of salt corresponds to about 0.4 g sodium, and 1 g sodium is equivalent to 2.5 g salt. One millimole

of sodium corresponds to 23 mg and is equivalent to about 58 mg sodium chloride. The aim of this scoping review is to describe the totality of evidence for the role of sodium for health-related outcomes as a basis for updating dietary reference values (DRVs) for the Nordic Nutrition Recommendations 2023 (NNR 2023) (Box 1). *Box 1.* Background papers for Nordic Nutrition Recommendations 2023

- This paper is one of many scoping reviews commissioned as part of the Nordic Nutrition Recommendations 2023 (NNR2023) project (Blomhoff et al, NNR report)
- The papers are included in the extended NNR2023 report but, for transparency, these scoping reviews are also published in Food & Nutrition Research
- The scoping reviews have been peer reviewed by independent experts in the research field according to the standard procedures of the journal
- The scoping reviews have also been subjected to public consultations (see report to be published by the NNR2023 project)
- The NNR2023 committee has served as the editorial board
- While these papers are a main fundament, the NNR2023 committee has the sole responsibility for setting dietary reference values in the NNR2023 project

Methods

This scoping review on sodium as salt follows the protocol developed within the NNR2023 project (1). The sources of evidence used in the chapter follow the eligibility criteria described previously (2). Seven qualified systematic reviews (SRs) on sodium were found (3). These include the recent scientific background to reference values for the USA (children 2015 and adults and children 2018), Australia and New Zealand (2017), European Food Safety Authority (EFSA 2019), and the systematic review by World Cancer Research Fund (WCRF) on preservation and processing (2018).

The search was conducted on 19.09.2021 on PubMed, and contained the following terms sodium [Title] AND ("2011"[Date - Publication]: "3000"[Date - Publication]) AND review [Publication Type] AND Humans[Filter] AND ("Diet" OR "Dietary" OR "Food" OR "Nutrition" OR "Nutritional"). The primary search was complemented by a new search on 31.1.2022.

Physiology

Sodium ion (Na⁺) is the major cation in the extracellular fluid (ECF). Together with the chloride anion (Cl⁻), they are the main contributors of the ECF osmolality (4). The ECF volume and the equilibrium between intracellular and extracellular osmolality is controlled by systems transporting sodium into the cell and by the energy-dependent sodium pump (Na⁺/K⁺-ATPase) that pumps sodium out of the cell in exchange for potassium. The differences in transmembrane distribution and activity gradients of sodium and potassium create a polarised cell membrane potential. Ion channels located in the cell membranes open in response to stimuli and allow ion flow and depolarisation of the cell membrane. The ion flow across the membranes is fundamental for the entry of water, glucose and amino acids into the cell, and for the electrical signalling of the nervous system, muscles, and heart (4).

The estimated body pool of sodium of an adult human is 1.3–1.5 g/kg of body weight (4), and half of it is located in ECF (2). Of the body total sodium, approximately 10.0% is in plasma, 30.0% in the interstitial fluid, 2.5% within cells, 2.5% in transcellular spaces, and 55.0% is bound in connective tissue, cartilage, bone, and soft tissues (5, 6). Approximately 98% of the ingested sodium is absorbed (7) and, on average, 93% of the ingested sodium is excreted in urine (8). Small amounts of sodium are lost in the faeces representing less than 2% of sodium intake (4). Dermal losses are influenced by sodium intake, sweat rate, hydration status and degree of heat acclimation, and represent the rest of the sodium losses (4). During profound sweating or in massive diarrhoea or vomiting, the extrarenal loss of sodium may become clinically important.

Renal sodium excretion is the difference between sodium filtered by the glomeruli and its total tubular reabsorption (5). Normally 98–99.99% of the filtered sodium is reabsorbed. The major mediators of sodium excretion are arterial blood pressure (BP) together with several neurohumoral factors. At the tubular level angiotensin II, aldosterone, norepinephrine and insulin stimulate sodium reabsorption. Dopamine, cAMP, cardiac natriuretic peptides, kinins, nitric oxide and some prostaglandins increase sodium excretion.

Assessment of nutrient status

There are no sensitive and specific biomarkers for estimation of sodium status of the body. The osmoregulatory systems control water intake and excretion, and keep the plasma sodium concentration in its normal range (135–145 mmol/L). Hyponatraemia may be caused by water dilution in the syndrome of inappropriate antidiuretic hormone excretion (SIADH) and is induced by some drugs and diseases, or when drinking large amounts of water in a short time (more than 1 l/h over several hours), by sodium and water retention in severe heart failure, and by true sodium depletion. Dietary causes for hyponatraemia are rare.

Sodium intake can be estimated by means of dietary questionnaires (24-h dietary recalls, food diaries or food frequency questionnaires [FFQ]) or by measuring 24-h urinary sodium excretion. There is a poor agreement with low correlations between sodium intake estimates based on dietary questionnaires and the gold standard 24-h urinary sodium excretion (9–11). Dietary questionnaires may underestimate sodium intake (9–11). In a pooled analyses

of five large validation studies, average underreporting of sodium intake was 28–39% for FFQ and 4–13% for dietary recalls. The FFQs were not specific for sodium intake, and they had limited ability to assess discretionary salt intake (9–11). At the individual level, higher 24-h urinary sodium excretion and higher body mass index (BMI) are strongly associated with underreporting of sodium intake, probably mostly as a consequence of underreporting of energy intake (9, 11).

Sodium intake varies day-by-day. Even at a constant intake the daily variation in sodium excretion is high, and, sodium excretion shows also an aldosteronedependent weekly rhythm (12–14). At least three but up to seven complete 24-h urine collections are needed to accurately estimate an individual's short-term sodium intake (12, 13). Multiple 24-h urine collections over several years are needed to reflect an individual's long-term sodium intake (7, 12). Complete collections of 24-h urinary sodium representing days of usual sodium intake (including weekend and week days) and possible seasonal variations is the gold standard to estimate average population sodium intake (7, 12).

Estimation of daily sodium intake from spot urine by using formulas based on age, urinary sodium, urinary creatinine, weight and age (Kawasaki), plus height (Tanaka), plus urinary potassium, and BMI instead of weight and height (Intersalt), has been used to assess cardiovascular risk in association with sodium intake (15-17). Correlations between daily sodium intake estimates from spot urine and 24-h urinary sodium are low and exhibit various sources of biases (7, 12). According to an analysis based on the Trials of Hypertension Prevention (TOHP) follow-up study, all three formulas overestimate at lower levels and underestimate at higher levels of sodium intake (17). The Kawasaki and Tanaka formulas overestimated on an average by 1,299 and 107 mg/day, respectively, and the Intersalt formula underestimated by 78 mg/day the average measured 24-h urinary sodium excretion over the trial period (17). The estimates of sodium intake by the three formulas were done by using sodium, creatinine, and potassium (Intersalt) concentrations measured from the 24-h urine. The results of the study suggest that sodium-independent variables in the formulas contribute to the biased estimates of sodium intake (17). One of them is urinary creatinine, which increases along with higher lean body mass (muscle mass), higher dietary protein intake, and higher levels of physical activity (18). BP level, secondary forms of hypertension, heart failure, medication and variation in the daily rhythm of sodium intake modify diurnal sodium excretion. This may further lead to serious systematic biases in the estimation of an individual's sodium intake from spot-urine and, consequently, in the estimation of relations between

sodium intake and cardiovascular disease (CVD). The J- or U-shaped relationships between sodium intake estimates and CVD disappear, and are linear, when repeated measurements of 24-h urinary sodium excretion is used in the assessment of sodium intake (17). Sodium intake estimated from spot-urine is not a valid method to assess sodium intake at the individual level, and cannot be used in the assessment of BP and cardiovascular risk in relation to sodium intake or in the comparisons of sodium intake across and within populations. (4, 7, 12, 19).

Dietary intake in the Nordic and Baltic countries

Approximately 60 to 70% of sodium in the diet comes from salt added to processed foods such as bread, bakery and other cereal products, meat and fish products, cheese, and ready meals such as pizza, pie, and soups (20–22). The contribution of sodium from added salt in cooking and at the table varies, but on average this intake constitutes approximately 10 to 20% of the total salt intake (20, 21).

Data on the dietary intake of sodium in Nordic and Baltic countries are based on the latest national dietary surveys done between the years 2007 and 2019, with 1,011 to 3,916 adult participants per survey, and with a participation rate ranging from 33 to 90%. Dietary data was collected by food records in Denmark (7-day) and Sweden (4-day), and with two separate 24-h dietary recalls in other countries (23). Daily mean sodium intakes were 4.4 g/day (11.0 g salt) in men and 3.2 g/day (8.0 g salt) in women in Denmark, 3.8 g in men (9.5 g salt) and 2.8 g (6.9 g salt) in women in Finland, 3.6 g/day (9 g salt) in men and 2.5 g/day (6.25 g salt) in women in Norway, 3.8 g (9.4 g salt) in men and 2.6 g (6.5 g salt) in women in Iceland, 3.6 g (9.0 g salt) in men and 2.7 g (6.9 g salt) in women in Sweden, 2.6 g (6.5 g salt) in men and 1.8 g (4.5 g salt) in women in Estonia, 3.2 g (7.9 g salt) in men and 2.4 g (6.0 g salt) in women in Lithuania, and 3.3 g (8.3 g salt) in men and 2.5 g (6.3 g salt) in women in Latvia (22, 23).

The contribution from discretionary salt intake - such as that from extra salt added to meals, etc. - is generally not included in these estimates. Data from Finnish population studies suggest that sodium intake assessed from 24 h and 48 h dietary recalls give similar estimates of the mean level of sodium intake as determinations of 24 h urinary sodium excretion (24). Salt intake was estimated based on food balance sheets, and the use of salt at the table was recorded (25). According to the latest Latvian dietary survey, 24-h urinary sodium excretions of 541 women and 470 men aged 19-64 years were 41 to 54% higher than daily sodium intake estimates made by means of dietary recalls, and were 5.1 g/day (12.8 salt) in men and 3.6 g/day (8.9 g salt) in women (25). Therefore, dietary estimates of population sodium intake should always be validated against 24-h urinary sodium excretion.

Health outcomes relevant to Nordic and Baltic countries

Deficiencies

Sodium deficiency does not normally exist in healthy individuals. Acute deficiency, hyponatraemia, can be caused by extreme and prolonged sweating in combination with large fluid intakes devoid of sodium or by massive vomiting and chronic diarrhoea. Severe adrenal insufficiency, excessive use of diuretics, sodium-losing kidney disease, and diabetic ketoacidosis may lead to sodium and water depletion. Clinical symptoms include: muscle seizures, loss of appetite, and circulation disturbances. Severe deficiency can result in coma and death.

Toxicities

Acute toxicity with fatal outcomes has been reported with single doses ranging from 6.8–10.2 g up to 400 g of sodium in adults and from 5 to 7 g in small children (26). The fatal outcomes were associated with severe hypernatremia and with plasma sodium levels ranging from 151 mmol/L to 255 mmol/L. A much smaller amount of an acute sodium load could be detrimental for subjects with heart failure, renal failure or decompensated liver cirrhosis. European Food Safety Authority defines hypernatremia as serum sodium concentration >145 mmol/L, and describes similar symptoms as hyponatraemia as well as headache, confusion, fever, nausea and vomiting (4).

Obesity and sodium

Obesity is associated with a higher sodium intake and increased sodium sensitivity (27–31). In a systematic review and meta-analyses of cross-sectional studies, a higher sodium intake was associated with overweight and obesity in adults and in children, and with abdominal obesity in adults (26). Insulin resistance, hyperinsulinemia, overactivity of the sympathetic nervous and renin-angiotensin systems are common features of obesity. Higher plasma insulin, aldosterone and cortisol levels enhance sodium tubular reabsorption and increase sodium load (28–31).

Hypertension

Elevated systolic BP was in 2017 the leading global risk factor for 10.4 million premature deaths and 218 million disability-adjusted life-years (32). Elevated BP increases the risk of cerebrovascular disease, ischemic heart disease, heart failure, peripheral artery disease, and end-stage renal disease. The risk of cardiovascular mortality rises without any threshold from BP levels less than 115/75 mmHg. The risk of fatal stroke is three-fold, and that of fatal myocardial infarction is two-fold, for a rise of 20 mmHg in systolic blood pressure (SBP) and 10 mmHg in diastolic blood pressure (DBP) (33). In the Framingham

follow-up study, the life-expectancies of 50-year-old men and women with a BP less than 120/80 mmHg were 5.1 and 4.9 years longer, respectively, as compared with men and women with SBP or DBP at least 140 mmHg and 90 mmHg (34). According the FinHealth 2017 health examination survey, the prevalence of hypertension, as determined by SBP at least 140 mmHg or DBP at least 90 mmHg or use of antihypertensive medication, increased along with age of men/women and was 28/11% in the age group of 30-39 years, 40/23% in the age group of 40-49 years, 62/47% in the age-group 50-59 years, 74/69% in the age-group of 60-69 years, 79/84% in the age-group of 70-79 years, and 80/87% in the age group of 80 years or more (35). Approximately 40% of the subjects treated for hypertension have their BP levels below 140/90 mmHg. Population-based BP data measured with exactly similar methods are difficult to find from other Nordic and Baltic countries. However, the existing data suggest that age-adjusted BP levels of men and women are quite similar in Nordic countries and somewhat higher in Baltic countries (36). Elevated BP also carries a high burden for increased cardiovascular morbidity and mortality in the Nordic and Baltic countries.

BP starts to rise in early childhood and leads most often to hypertension later in life as a consequence of functional and structural microvascular and macrovascular alterations. Hypertension is characterised by rarefication of the capillary density, increased contractility of the resistance arteries, stiffening of the large conduit arteries, and changes in the structure and function of the heart. Changes in the microvasculature lead to increased peripheral resistance. Increased fibrosis and decreased elastin to collagen ratio in the arterial wall of large conduit arteries (aorta) lead to their stiffening and to changes in the pulsatile components of the pulse wave, clinically observed by changes (peaking) in the central pulse-wave form, increased aortic pulse wave velocity, and, robustly, by increased pulse pressure (difference between SBP and DBP) and SBP. High intake of sodium plays a significant role in the long-run process of the development of hypertension (37–40). Increased intracellular calcium in vascular smooth muscle cells, induced primarily by high sodium intake and sodium retention, may be one of the key factors behind increased contractility of resistance arteries, impaired vascular endothelial function, and increased peripheral resistance in sodium-sensitive hypertension (41, 42). High sodium intake may induce stiffening of the large arteries and structural changes of the left ventricle also by pressure-independent mechanisms (41, 43).

Adults

Observational studies have shown that hypertension is rare and that BP does not normally rise with age in

populations with a very low sodium intake (<0.8 g/day) (44). In the Intersalt study, the relationship between 24-h sodium and potassium excretion and BP was investigated. The study included 10,000 men and women aged 20-59 years from 52 centres around the world. The median urinary sodium excretion varied from 0.9 mmol/day (0.02 g/day) to 242 mmol/day (5.57 g/day) between the centres. In four centres (Brazil: Xingu and Yanomamo; Papua New Guinea; Kenya) with low sodium excretion (0.02-1.31 g/day, lowest in the two centres in Brazil and highest in Kenva), the BP levels were low and practically no hypertension was observed. Across the four centres, the average BP increased from 96.0/60.6 mmHg to 113.3/66.0 mmHg along with increasing sodium excretion. No age-related increase in BP was observed in the two centres with the lowest sodium excretion (Xingu and Yanomamo), and the age-related rise was only small in the other two centres (Papua New Guinea and Kenya). According to within-population analyses of all 52 centres, a 100 mmol (4.35 g) higher 24-h urinary sodium was associated with a 3/0 to 6/3 mmHg higher SBP/DBP (analyses with and without BMI (45). In cross-population analyses, a 100 mmol higher 24-h urinary sodium was associated with a 5-7/2-4 mmHg higher SBP/DBP (45) and with an estimated mean difference of 10-11/6 mmHg in SBP/DBP between 55 and 25 years of age. Increased BMI and heavy alcohol intake were strongly related to increased BP. A positive linear correlation between sodium intake and BP has been confirmed in other cross-sectional studies and meta-analyses of observational studies in adults as well as in children and adolescents (46-48). A positive nonlinear association was observed in a large study where daily sodium intake was estimated from spot-urine (49).

Several meta-analyses of clinical trials of dietary salt reduction have been published, and they differ in scope and inclusion criteria. The meta-analyses of Agency for Healthcare Research and Quality (AHRQ) in the United States of America (US) identified 47 comparisons that met their inclusion criteria (50). The analyses were based on parallel randomised controlled trials (RCTs) and on crossover RCTs with a washout period of at least 2 weeks. The net differences in sodium intakes between high and low sodium groups or phases ranged from 0 to 120 mmol/ day (0 to 2,760 mg/day) with a weighted mean difference of 42 mmol/day (966 mg/day). BP measurements in the included studies were made in sitting or supine position (the intervention effect on supine and sitting BP may differ from each other). Several studies with hypertensive subjects using BP-lowering medication, also diuretics, were included. A decrease in BP-lowering medication was one of the primary endpoints in one study showing halving of BP-medication during sodium restriction. Yet, the net BP-changes between the sodium lowering and the usual sodium intake arms were used for the analyses. There were also errors in numerical values of the net BP changes between the sodium restriction and control groups. In the analyses of all trials with hypertensive and normotensive participants combined, sodium reduction decreased SBP by 3.2 mmHg (95% confidence interval [CI]: -4.1 to -2.4mmHg) and DBP by 2.3 mmHg (95% CI: -2.9 to -1.6mmHg). SBP decreased by 4.1 mmHg (95% CI: -5.2to -3.1 mmHg) and DBP by 2.6 mmHg (95% CI: -3.3to -1.9 mmHg) in hypertensive participants. In normotensive participants, SBP decreased by 1.5 mmHg (95% CI: -2.8 to -0.3 mmHg) but DBP did not decrease significantly (-0.6 mmHg; 95% CI: -1.3 to +0.1 mmHg). A larger difference in sodium intakes between high and low sodium groups or phases and a higher mean age was associated with a larger BP decrease.

The report of the EFSA Panel on Nutrition, Novel Foods and Food Allergens identified 32 RCTs and 35 comparisons that met their inclusion criteria (4). The analyses were based on seven parallel RCTs and 25 crossover trials. The net differences in sodium intakes between high and low sodium groups or phases ranged from 13 to 285 mmol/day (300 to 6,560 mg/day), with a median value of 72 mmol/day (1,656 mg/day). In the analyses of all trials with hypertensive and normotensive participants combined, sodium reduction decreased SBP by 3.9 mmHg (95% CI: -5.1 to -3.1 mmHg) and DBP by 2.0 mmHg (95% CI: -2.8 to -1.2 mmHg). A larger effect was found in hypertensive than normotensive subjects. SBP decreased by 5.6 mmHg (95% CI: -8.1 to -3.1 mmHg) and DBP by 2.9 mmHg (95% CI: -4.2 to -1.6 mmHg) in hypertensive participants. In normotensive participants, SBP decreased by 2.0 mmHg (95% CI: -3.3 to -0.7 mmHg) and DBP by 0.9 mmHg 95% CI: -1.6 to -0.2 mmHg). According to the dose-response analyses of eligible trials, for each 100 mmol /day (2.3 g/day) increase in mean 24-h urinary sodium excretion, mean SBP increased by 5.3 mmHg (94% CI: 3.6 to 6.9 mmHg) and mean DBP increased by 2.6 mmHg (95% CI: 1.6-3.7 mmHg).

Graudal et al. studied effects of low versus high sodium diets on BP, renin, aldosterone, catecholamines, and serum lipids (51). The Cochrane Database systematic review included 195 intervention studies published between 1973 and 2018, 12,296 individuals, and sodium restriction durations of 3 to 1,100 days. The effects on BP were shown separately for white, black and Asian participants. A mean sodium reduction from 203 mmol/ day (4,670 mg/day) to 65 mmol/day (1,490 mg/day) decreased SBP in hypertensive white participants by 5.7 mmHg (95% CI: -6.7 to -4.7 mmHg; 3,998 participants, 88 trials) and DBP by 2.9 mmHg (95% CI: -3.4 to -2.3 mmHg; 4,032 participants, 89 trials). In normotensive white participants SBP decreased by 1.1 mmHg (95%) CI: -1.7 to -0.6 mmHg; 5,982 participants, 95 trials) but DBP was unchanged (+0.1 mmHg; 95% CI: -0.4 to +0.4 mmHg; 6,276 participants, 96 trials). The effects on BP were largely similar in black and Asian participants. During sodium restriction, serum cholesterol increased by 0.08 mmol/l (95% CI: 0.04 to 0.13 mmol/l; 917 participants in 27 trials), triglycerides increased by 0.13 mmol/l (95% CI: 0.05 to 0.21 mmol/l; 712 participants in 20 trials), LDL-cholesterol tended to increase (0.06 mmol/l; 95% CI: -0.03 to 0.15 mmol/l), but no significant changes were seen in HDL-cholesterol. During low sodium intake, renin (2,904 participants, 82 trials), aldosterone (2,506 participants, 66 trials) and noradrenalin (878 participants, 35 trials) were higher than during high sodium intake. A weakness of the analysis of trials on normotensives and of studies on the sodium and BP regulating hormones was their short duration (usually not more than 14 days, often 3-7 days). Most of the studies were experimental and explored sodium and BP regulating systems, sodium sensitivity and BP reactivity in healthy, young voluntary subjects during a very low sodium intake of 20-30 mmol/ day (0.5-0.7 g/day) and during a high sodium intake of usually 200-300 mmol/day (4.6-6.9 g/day), but up to 800 mmol/day (18.4 g/day). The changes in renin, aldosterone and noradrenaline levels can be estimated to be physiological under these circumstances. Furthermore, included studies exploring effects of sodium intake on BP reactivity showed that BP rise to noradrenaline infusion was smaller during strict sodium restriction than after a heavy sodium load (52, 53).

The meta-analyses of Aburto et al. included randomised controlled studies that had a duration of at least 4 weeks, achieved a difference of at least 40 mmol/day (0.9 g/day) in sodium intake between intervention and control and measured 24-h urinary sodium excretion (54). They identified 36 studies with 49 comparisons, and found that a reduction in sodium intake reduced SBP by 3.4 mmHg (95% CI: -2.5 to -4.3 mmHg) and DBP by 1.5 mmHg (95% CI: -1.0 to -2.1 mmHg). The meta-analyses of three comparisons showed that SBP decreased by 3.5 mmHg (95% CI: -0.8 to -3.1 mmHg) and DBP by 1.8 mmHg (95% CI: -0.5 to -3.1 mmHg) when the achieved sodium intake of the intervention was less than 2 g/day as compared to at least 2 g/day. In the meta-analyses of 11 studies (2,339 participants) that reported changes in blood lipids, reduced sodium intake had no effects on total cholesterol (0.02 mmol/L; 95% CI: -0.03 to 0.07 mmol/l), LDL-cholesterol (0.03 mmol/l; 95% CI: -0.02 to 0.08 mmol/l), HDL-cholesterol (-0.01 mmol/L; 95% CI: -0.03 to 0.00 mmol/L) or triglycerides (0.04 mmol/L; 95% CI: -0.04 to 0.09 mmol/L). The four randomised trials reporting plasma adrenalin and seven trials reporting noradrenaline detected no effects of sodium restriction on plasma catecholamine levels.

The dose-response meta-analyses of Filippini et al. included 85 experimental studies with at least 4 weeks of

follow-up, 24-h urinary sodium excretion measurements, dietary sodium changes by diet or supplementation, and measurements of SBP and DBP (55). The study reports were published between 1973 and 2018, and the analyses included over 10,000 individuals. The achieved difference in 24-h urinary sodium between the intervention and control groups ranged from 5 to 309 mmol/day (0.1-7.1 g/day) with a median difference of 80 mmol/day (1.8 g/day). According to linear regression analyses, every 100 mmol/ day (2.3 g/day) reduction in urinary sodium excretion was associated with a lower SBP of 5.6 mmHg (95% CI: -6.6 to -4.5 mmHg) and a 2.3 mmHg lower DBP (95% CI: -3.3 to -1.7 mmHg). Correspondingly, every 1 g/day decrease in sodium excretion was associated with a lower SBP of 2.4 mmHg (95% CI: -2.9 to -2.0 mmHg) and a 1.0 mmHg lower DBP (95% CI: -1.3 to -0.7 mmHg). A dose-response relationship was observed with a mean SBP/DBP reduction of 6.5 mmHg (95% CI: -7.8 to -5.2 mmHg)/3.0 mmHg (95% CI: -3.7 to -2.3 mmHg) among hypertensives and a reduction of 2.3 mmHg (95%) CI: -3.3 to -1.3 mmHg)/0.8 mmHg (95% CI: -1.9 to +0.3 mmHg) among normotensives per 100 mmol/day (2.3 g/day) reduction in sodium excretion. The approximately linear relationship between achieved sodium intake and mean SBP and DBP extended down to sodium excretion of 1.5 g/day. As compared to sodium excretion of 2.0 g/day, sodium excretion of 1.5 g/day was associated with a lower SBP of 1.3 mmHg (95% CI: -1.8 to -0.8 mmHg) and a lower DBP of 0.4 mmHg (95% CI: -0.8 to -0.0 mmHg). Correspondingly, an increase in sodium excretion from 2 to 2.5 g/day was associated with a higher SBP of 1.3 mmHg (95% CI: +0.6 to +1.7 mmHg) and a higher DBP of 0.4 mmHg (95% CI: +0.1 to +0.6 mmHg).

Only a few studies have examined the long-term effects on BP of sodium restriction. Jula et al. (56) studied the effects on BP, serum lipids and echocardiographically assessed heart structure and function of a non-pharmacological treatment based mainly on sodium restriction in a 12-month controlled randomised study with 91 middle-aged untreated and mildly hypertensive men and women. The estimated daily sodium intakes, calculated from 24 h urine samples, decreased in men from 227 mmol/day (5.2 g/day) to a mean level of 105 mmol/day (2.4 g/day), and in women from 129 mmol/day (3.0 g/day) to a mean level of 63 mmol/day (1.4 g/day). After 12 months non-pharmacological treatment, the mean weight in men was 1.9 kg lower and in women 0.3 kg higher compared to baseline. In the treatment group, energy derived from fat decreased in men by 4% and in women by 3% reflecting decreased intake of saturated and monounsaturated fats. The net BP decrease (the difference in changes between the treatment and control groups) during the 12 months was 8.2 mmHg for SBP and 5.8 mmHg for DBP in men and 9.5 mmHg for SBP and 5.6 mmHg for DBP in women. All changes were significant. In the treatment group, LDL-cholesterol decreased by 6.8% in men and by 12.1% in women. The non-pharmacological treatment decreased left ventricular mass (LVM) by 5.4%, and the decrease was larger (8.6%) in those subjects with baseline LVM/body surface area above the median values of both men and women (57). Regression of left ventricular hypertrophy was associated with decreased plasma levels of atrial natriuretic peptide (ANP), indicating decreased load of the heart (58). The study illustrated that a 1-year nonpharmacological treatment with reduced saturated fat intake partly compensated by increased unsaturated fat intake together with sodium restriction down to 2.4 g/ day in men and down to 1.5 g/day in women, decreased statistically and clinically significantly both BP and LDLcholesterol, and had favourable effects on the left ventricle of the heart.

In the Dietary Approaches to Stop Hypertension (DASH) trials, the effects of various controlled diets on the BP of adult Americans with normal or moderately elevated BP were studied (59, 60). The study assessed the influence of sodium intake on BP in 412 subjects and on fasting serum lipids in 390 subjects who were randomly assigned to follow either a control diet typical of the sodium intake in the US or to follow the DASH diet (60, 61). In both groups, a second randomisation was done, and the subjects followed their assigned diets at three sodium levels for 30 days in random order in a crossover design. The subjects were selected among adults 22 years or older who were not taking antihypertensive medication and who had a SBP ranging from 120 mmHg to 160 mmHg and a DBP ranging from 80 mmHg to 95 mmHg. The control diet had a fat composition corresponding to the usual American diet (36 E% total fat, 14 E% saturated fat), but a low content of fruits, vegetables, and dairy products. The DASH diet was rich in fruits, vegetables, and low-fat dairy products, but low in edible fats, snacks, and sweets, and had a low content of total fat (25 E%), saturated fat (7 E%), and cholesterol. The content of calcium, potassium, and magnesium in the control diet was lower than in the average US diet, but was higher in the DASH diet. The intake of dietary fibre was similar in both groups. Within the assigned diets, sodium levels were adjusted to provide a daily intake of 150 mmol (3.5 g, defined as high), 100 mmol (2.3 g, defined as intermediate), and 50 mmol (1.2 g, defined as low) for 30 consecutive days each, in random order. The average sodium intake, calculated from 24 h urine samples, was 141-144 mmol/day (3.3 g/ day) during the high sodium phase, 64-67 mmol/day (1.5 g/day) during the low sodium phase and 106-107 mmol/ day (2.4 g/day) during the intermediate sodium phase. Reduction of sodium intake from the high to the intermediate level significantly decreased SBP by 2.1 mmHg during the control diet and by 1.3 mmHg during the

DASH diet. A further reduction from the intermediate to the low level caused additional reductions of 4.6 mmHg during the control diet and 1.7 mmHg during the DASH diet. A regression analysis of these data showed that a reduction in the sodium intake of 100 mmol(2.3 g) per day would decrease SBP by 3 mmHg in the DASH diet group and by 7 mmHg in the control diet group. Corresponding values for DBP were 1.5-2 mmHg and about 3 mmHg, respectively. The effects of sodium were observed in normotensive and hypertensive subjects, in subjects of different ethnicities, and in both women and men. The effects were not dependent on weight, but were higher in older (>45 years) than in younger subjects (59, 60). Total cholesterol was 0.41 mmol/l (7.8%) and LDL-cholesterol 0.33 mmol/l (9.7%) lower during the DASH diet than during the control diet (61). No differences were seen in fasting blood lipids across the three sodium intake phases (61).

The DASH trials clearly showed a BP lowering effect of sodium restriction, ranging from 1.5 to 2.4 g/day and finally to 3.5 g/day, that was independent of other dietary and lifestyle factors. An important finding was that the BP reduction was larger in the control group than in the DASH diet group. This implies that the benefits of sodium restriction are more pronounced among persons consuming a diet that is less than optimal in terms of fat, fruit, vegetable and low-fat dairy food intake (which is similar to the current dietary patterns in the Nordic countries) than among those already consuming a diet in line with the general nutrition recommendations (23). It should also be noted that the estimated daily potassium intakes, calculated from 24 h urine samples, indicated a potassium intake of 40-42 mmol/day (1.6 g/day) in the control group and 75-81 mmol/day (2.9-3.2 g/day) in the DASH diet group during the high, intermediate, and low sodium intake phases. It is also noteworthy that the mean daily dietary intake of potassium in the Nordic and Baltic countries is approximately twice as high as that of the DASH control group, and even somewhat higher than that of the DASH diet group (23). In the DASH trial, an interaction between dietary sodium and potassium may explain why the BP-lowering effect of sodium was higher during the control diet with a low potassium intake and lower, but still clinically significant, during the DASH diet with a high potassium intake. Sodium restriction from 3.3 to 2.4 g/day and down to 1.5 g/day had no effects on blood lipids. A limitation of the study was the relatively short duration (30 days) of the sodium restriction phases.

Children and adolescents

The BP of children living in industrialised countries rises with age, and the increase is more rapid in children of hypertensive parents than in children of normotensive parents (62–65). In the Finnish Special Turku Coronary Risk Factor Intervention Project (STRIP) study (65), SBP of children living in southwestern Finland increased with age along with sodium intake. At the age of 15 years, their SBP exceeded by 15–20 mmHg that of the adults in populations consuming extremely low-sodium diets (Brazil: Xingu and Yanomamo), and was 114 mmHg in girls and 121 mmHg in boys. The mean daily sodium intake was 1.5 g at the age of 13 months and 3.0 g at the age of 15 years. Similar levels of salt intake by children have been reported in other countries (66, 67). Childhood BP tracks with adult BP (68, 69) and predicts early signs of atherosclerosis in adolescence (70) and adulthood (70, 71).

A randomised trial among 476 Dutch newborn infants studied the effect of a low-sodium (average 120 mg/day) or a normal-sodium (average 330 mg/day) diet on BP during the first 6 months of life (72). The sodium intake in the low-sodium group was approximately the same as the intake of breastfed infants, whereas the intake in the normal-sodium group was similar to the sodium intake of infants fed commercial infant formula. At the end of the trial, SBP in the low-sodium group was 2.1 mmHg lower than in the control group. The authors also measured BP in 167 children from the original cohort (35% of the original study participants) after 15 years of follow-up. The adjusted SBP at follow-up was 3.6 mmHg lower and the DBP was 2.2 mmHg lower in adolescents who as infants had been assigned to the low-sodium group compared with those assigned to the control group.

Three meta-analyses of controlled trials have assessed effects of reduced salt intake on BP in children and adolescents (48, 54, 73). He and MacGregor identified 10 trials with 966 participants (73). The mean ages of the study participants ranged from 8 to 16 years, and the median duration of the trials was 4 weeks. A reduction in salt intake by 42% decreased SBP by 1.2 mmHg (95% CI: -0.6 to -1.8 mmHg) and DBP by 1.3 mmHg (95% CI: -0.7 to -1.9 mmHg). In the analyses of three trials with infants and with a median duration of 20 weeks, a reduction in sodium excretion by 54% decreased SBP by 2.5 mmHg (95% CI: -0.9 to -4.0 mmHg). Aburto et al. identified nine studies with 1,299 participants aged 2.6 to 19.8 years, and found that a reduction in sodium intake reduces SBP by 0.8 mmHg (95% CI: -0.3 to -1.4 mmHg) and DBP by 0.9 mmHg (95% CI: -0.1 to -1.2 mmHg) (54). Leyvraz et al. included 13 studies with 1,687 participants and with a mean duration of 16 weeks, and found that 1.2 g/day reduction in daily sodium intake decreased SBP by 0.6 mm Hg (95% CI: -0.5 to -0.8 mmHg) and DBP by 1.2 mmHg (95% CI: -0.4 to -1.9 mmHg) (48).

Cardiovascular diseases

A number of prospective cohort studies, systematic reviews and meta-analyses have investigated the association between dietary salt intake and the risk of stroke and other cardiovascular events. Sodium intake has been estimated in the studies by dietary questionnaires (24-h dietary recalls, food diaries, FFQs), by measuring 24-h urinary sodium excretion or by spot urine samples and estimation equations. Twenty-four-hour urinary sodium is the gold standard method to estimate sodium intake. Because of a large daily variation in sodium excretion, multiple measurements over a longer time is needed to assess an individual's usual longer term sodium intake (see above).

The comparative effectiveness review of AHRQ assessed the relation of sodium intake with CVD, CVD morbidity, CVD mortality, and total mortality (19). They concluded that sodium levels appear to be associated with risk of all-cause mortality, but the shape of the relationship could not be determined because of insufficient source of error. They concluded also that evidence was insufficient to assess association of sodium level and risk for CVD, coronary heart disease (CHD), or stroke morbidity and mortality. The assessment was based on 13 prospective cohort studies (15 cohorts) and one nested case-control study. Sodium intake was assessed by 24-h urinary sodium excretion in five cohorts, by overnight sodium excretion in one cohort, by spot urine samples in five cohorts, by 24-h dietary recall in three cohorts, and by 3-day food records in one cohort. It should be noted that the cohort studies using imprecise assessment methods of sodium intake carry a high risk of bias and are not suitable for assessments of CVD risk in relation to sodium intake. According to random-effects model meta-analyses of four studies with measured 24-h urinary excretion, the adjusted risks of total mortality increased (relative risk [RR]: 1.09; 95% CI: 1.00 to 1.19) per 50 mmol/l (1,150 mg/day) increase in urinary sodium excretion in generally healthy populations. One of the five studies that measured 24-h urinary sodium excretion found J-shape associations between sodium intake and mortality, and negative associations between higher sodium intake and CVD mortality. The prospective study was based on two cohorts, the Flemish Study on Genes, Environment and Health Outcomes (FLEMENGHO) and the European Project on Genes in Hypertension (EPOGH) (74). The outcome cohort consisted of 3,681 participants (2,674 in the FLEMENGHO and 1,007 in the EPOGH cohort). During the median follow-up time of 7.9 years, 84 fatal (81 in the FLEMENGHO and three in the EPOGH cohort) and 134 nonfatal outcomes (113 in the FLEMENGHO and 21 in the EPOGH cohort) were registered. Pulmonary embolism (n = 11) and pulmonary heart disease (n = 14) were unorthodoxly included as cardiovascular outcomes. The number of fatal (n = 29)and nonfatal (n = 43) heart failures was high, as 52 participants with CVD at baseline were excluded from the outcome cohort. Men and women in the lowest 24-h urinary sodium excretion tertile were slightly older and had lower body weight than men and women in the medium and high tertiles. Lower urinary volume and creatinine excretion suggest misclassification of participants and/ or selection of more fragile subjects with lower muscle mass into the lowest sodium excretion tertile. The risk of bias in the FLEMENGHO and EPOGH cohort was unexpectedly estimated as medium instead of high by the AHRQ.

A systematic review and meta-analysis of prospective studies published between 1996 and 2008 assessed the relation between habitual salt intake and stroke or total cardiovascular events (75). The analysis included 19 independent cohorts from 13 studies with a total of 177,025 participants from the US, Finland, Japan, the Netherlands, Scotland and Taiwan, and over 11,000 vascular events. Follow-up varied between 3.5 and 19.0 years. Sodium intake was assessed by 24-h dietary recall (n = 4), FFQ (n = 3), FFQ and overnight urinary sodium excretion (n = 1), 24-h urinary sodium excretion (n = 4), or by a questionnaire (n = 1). A higher estimated sodium intake of 100 mmol/day (2.3 g/day) was associated with a higher incidence of stroke (pooled relative risk 1.23; 95% CI: 1.06 to 1.43) and CVD (RR: 1.14; 95% CI: 0.99 to 1.32). Higher sodium intake was associated significantly with higher CVD risk (RR: 1.17, 95% CI: 1.02 to 1.34) when one study with 24-h urine samples collected 5 days after the subjects had been asked to avoid consumption of foods with a high salt content was not included in the analyses (76).

The systematic review and meta-analyses of Aburto et al. included 12 prospective cohort studies (54). Sodium intake was assessed by 24-h dietary recall (n = 3), by FFQ (n = 3), 24-h urinary sodium excretion (n = 4), overnight urinary sodium excretion (n = 1), or was estimated from spot urine samples with Kawasaki equation (n = 1). The meta-analyses of three studies and five comparisons find an increased risk of CHD mortality with higher sodium intake (RR: 1.32, 95% CI: 1.13 to 1.53) (41). Meta-analyses of 10 studies detected an increased risk of stroke (RR: 1.24, 95% CI: 1.08 to 1.43) and an increased risk of fatal stroke (RR: 1.63, 95% CI: 1.27 to 2.10) along with higher sodium intake. The meta-analyses of sodium intake and CVD and CHD were inconclusive.

Poggio et al. performed a systematic review and meta-analyses of 11 studies and 229,785 participants representing the general population (77). The average follow-up time was 13.4 years (range 5.5 to 19.0 years). Sodium intake was assessed by 24-h dietary recall (n = 3), FFQ (n = 3), FFQ and 3-day dietary record (n = 1), or by 24-h urinary sodium excretion (n = 5). Higher sodium intake was associated with higher cardiovascular mortality

(RR: 1.12; 95% CI: 1.06 to 1.19). The meta-regression analyses showed that CVD mortality increased by 1% (P = 0.016) for every increase of 10 mmol/day (230 mg/day) in sodium intake.

Wang et al. assessed the relation between sodium intake and CVD with a systematic review and meta-analysis of 36 studies, 616,905 participants, and a follow-up time ranging from 2.7 to 29 years (78). Sodium intake was assessed by 24-h dietary recall (n = 8), FFQ (n = 7), dietary records (n = 2), self-administered questionnaire (n = 1), 24-h urinary sodium excretion (n = 18), or by spot urine sodium excretion (n = 1). Higher sodium intake was associated with higher adjusted risk of CVD (RR: 1.19; 95% CI: 1.08 to 1.30). According to dose-response analyses of 20 studies, the risk of CVD increased linearly up to 6% for every 1 g increase in dietary sodium intake.

Ma et al. assessed relations of sodium intake and cardiovascular risk by including individual-participant data from six prospective cohorts of generally healthy subjects (79). Sodium intake was assessed with the use of at least two 24-h urine samples per participant in six different cohorts including 10,709 participants with a mean age of 51.5 years. During the median follow-up time of 8.8 years, 571 cardiovascular events (coronary revascularisation, fatal or nonfatal myocardial infarction, fatal or nonfatal stroke) were ascertained. Higher sodium excretion, lower potassium excretion and higher sodium/potassium ratio were associated with a higher cardiovascular event risk in analyses adjusted for all confounding factors. In analyses comparing the highest sodium excretion quartile (median 4.7 g/day) with the lowest quartile (median 2.2 g/day), higher sodium excretion was associated with an increased risk of cardiovascular event (hazard ratio [HR] 1.60; 95% CI: 1.19 to 2.14). The spline analyses of the data showed a linear association over the range of sodium excretion. According to dose-response analyses, the risk of a cardiovascular event increased by 18% for every 1 g increase in daily sodium excretion (HR: 1.18; 95% CI: 1.08 to 1.29).

Prospective studies based on a single daily sodium intake estimate from spot urine of the participants have shown J-shaped association between estimated sodium intake and CVD (15, 16). Sodium intake estimates in these studies have been made from morning spot urine samples by using the Kawasaki formula, which at the individual level underestimates high and overestimates low sodium intake and at the study population level overestimates average sodium intake (17).

The observational analyses of two large randomised drug-treatment studies, ONTARGET (Ongoing Telmisartan Alone and in combination with Ramipril Global Endpoint Trial) and TRANSCEND (Telmisartan Randomized Assessment Study in ACE Intolerant Subjects with CVD), included two cohorts with 28,880 patients from 733 centres in 40 countries (15). Both trials included patients with a high risk of CVD (age \geq 55 years with established CVD or high-risk diabetes). According to medical history, 48% of the study population had suffered from a previous myocardial infarction, 21% from a stroke, 70% had hypertension, and 37% had diabetes. The mean follow-up time was 56 months. The main outcome measures were cardiovascular death, myocardial infarction, stroke and hospitalisation for congestive heart failure. Subjects were divided into seven groups according to their estimated baseline sodium intake (<2 g/day (n = 818), 2.00–2.99 g/day (n = 2,654), 3.00-3.99 g/day (n = 5,699), 4.00-5.99 g/day(n = 14, 156), 6.00-6.99 g/day (n = 3, 380), 7.00-8.00 g/day(n = 1,326), and >8.00 g/day (n = 847). The use of diuretics, the history of hypertension and diabetes, and the prevalence of atrial fibrillation was highest among the groups with an estimated sodium intake of <2.0 and >8.0 g/day and lowest in the reference group with an estimated sodium intake of 4.00-5.99 g/day. Compared with a baseline sodium excretion of 4.00 to 5.99 g/day, sodium excretion greater than 7.0 g/day was associated with an increased risk of myocardial infarction, stroke, cardiovascular death, and congestive heart failure. Sodium excretion of less than 3.0 g/day was associated with an increased risk of congestive heart failure and cardiovascular death, but not with an increased risk of myocardial infarction or stroke. Reverse causality and inaccurate estimates of the participant's sodium intake are the plausible explanations for the observed J-shape associations between sodium intake, congestive heart failure and cardiovascular death.

The Prospective Urban Rural Epidemiology (PURE) study enrolled and followed 156,424 subjects, aged 35 to 70 years, living in 628 urban and rural communities in 17 countries (16). A morning fasting urine sample was obtained and daily sodium intake was estimated by using the Kawasaki formula. The mean follow-up time was 3.7 years. The primary outcome was defined as death from any cause or a cardiovascular event (cardiovascular death, stroke, myocardial infarction or heart failure). The study included 101,945 participants of whom 42% were from China. The mean estimated daily sodium intake was 4.9 g. A primary outcome occurred in 3,317 participants. As compared with the reference group with an estimated sodium intake of 4.00-5.99 g/day, a sodium intake of \geq 7.0 g/day was associated with increased risk of the primary outcome and of its components. Respectively, an estimated sodium intake of <3.0 g/day was associated with an increased risk of the primary outcome. As compared with the reference group, subjects in the low sodium group were older, had a higher proportion of women, subjects using cardiovascular medication or having a history of CVD. Reverse causality and selection bias into different salt groups by the inaccurate estimation method of individual sodium intake are the plausible explanators of the U-shaped association of sodium intake and cardiovascular events in the PURE study.

The effects of predictive formulas on the estimation of sodium intake and the associations of these estimates with mortality was analysed using TOHP follow-up data (17). The analyses included 2,974 individuals aged 30 to 54 years who were not assigned to sodium intervention. During the follow-up of 24 years, 272 deaths occurred. Sodium intake was assessed from three to seven 24-h urinary sodium collections by direct measurements of 24-h sodium excretion and by three sodium estimation formulas. As compared to measured average 24-h urinary sodium of 3.8 g/day over the trial, the estimated average 24-h urinary sodium based on the Kawasaki formula overestimated sodium intake by 1.3 g/day. There was a significant linear relationship between the average measured daily sodium excretion and all-cause mortality. The relationships between average sodium intake and mortality were J- or U-shaped when sodium intake was estimated by using the spot-urine-based formulas of Kawasaki, Tanaka or Intersalt. The study shows clearly that even average sodium intake estimated from urine samples over a long time by using spot-urine-based formulas is not a valid method to assess health risks in association with sodium intake. The study confirms the findings of other studies (79, 80) that 24-h urinary sodium averaged over several measurements and longer time is the gold standard in the estimation of an individual's usual sodium intake and the health risks associated with sodium intake.

Intervention studies

A recent cluster-RCT involving 20,995 persons from 600 villages in rural China studied the effect of potassium enriched salt substitute on CVD (81). The mean age of the participants was 65.4 years, 73% of the participants had a history of stroke, and 88% had a history of hypertension. Twenty-four-hour urine samples were collected from a representative sample of study subjects at baseline and at 12-month intervals during the study. During the mean follow-up time of 4.7 years, 4,172 deaths occurred. The baseline sodium and potassium intakes were 4.3 and 1.4 g/day, respectively. During the follow-up the, the average sodium intake was 0.35 g/day lower, the potassium intake 0.80 g/day higher, and the average SBP 3.3 mmHg lower in the salt substitute group. Compared to controls, significantly less strokes (RR: 0.86; 95% CI: 0.77 to 0.96), major cardiovascular events (RR: 0.86; 95% CI: 0.80 to 0.94), and deaths from any causes (RR: 0.88; 95% CI: 0.82 to 0.95) occurred in the salt substitute group.

Population follow-up data ('natural non-randomised interventions') have also shown that BP and CVD mortality decreases along with decreased sodium intake and other favourable dietary changes. In Finland, salt intake decreased by 40% from the 1970s to 2002 along with a decreased intake of saturated fats and increased intake of fruits and vegetables (82–84). The dietary changes were associated with a 10–20 mmHg and 6–10 mmHg decrease in population SBP and DBP, respectively, and with a 70% decrease in stroke and CHD mortality (85, 86). The observed decrease in BP was much larger than expected from cross-sectional and trial data. This suggest that the effect of dietary sodium on the vasculature and BP rise accumulates with age, and that the current estimates underreport the effects on BP of a long-term population-based prevention with decreased sodium intake.

Other health outcomes

Sodium reduction decreases albuminuria and proteinuria, markers of kidney glomerular damage. The metaanalysis of Nomura et al. included 20 studies with 766 participants and a mean study duration of 14 days (range 4 to186 days) (87). The renal outcomes (albuminuria, albumin-to-creatinine ratio, proteinuria, protein-to creatinine ratio) differed between the studies. The pooled effects size showed a 53% (95% CI: 21 to 84%) higher albumin or protein excretion on the high-sodium than on the lowsodium diet. The meta-analysis of D'Elia et al. included 11 studies, 23 cohorts and 516 participants with a follow-up time of 1-6 weeks (88). An average reduction in sodium intake of 92 mmol/day (2.1 g/day) was associated with a 32% (95% CI: -44 to -19%) reduction in urinary albumin excretion. The effect was higher in cohorts using renin-angiotensin-aldosterone-blocking medication, in studies with a duration of at least 2 weeks, and in patients with evidence of kidney damage. Sufficiently powered long-term intervention studies are needed to assess if sodium restriction can prevent glomerular damages in high-risk populations, and, slow down worsening of kidney function and development of end stage renal disease in subjects with kidney disease.

Several studies have indicated a positive relationship between sodium excretion and calcium excretion and that sodium intake might play a role in the aetiology of osteoporosis and kidney stones (89, 90).

Two prospective cohort studies have assessed the risk between sodium intake and diabetes (91, 92). The first study included 1,935 men and women, aged 35–64 years from the WHO FINMONICA 1982 and 1987 cohorts who had complete data on 24-h urinary sodium (91). During the mean follow-up time of 18.1 years, 129 incident cases of type 2 diabetes occurred. The multivariate adjusted HR for diabetes in the highest (>6.0 g/day in men and >4.5 g/day in women) versus lowest quartile (<3.7 g in men and <2.8 g/day in women) of 24-h urinary sodium was 2.05 (95% CI: 1.43 to 2.96). In the second study, 4,630 men and women aged 25–64 years collected 24-h urine samples between 1979 and 2002,

and were followed up to 14 years (92). The study subjects of the population-based samples comprised of participants of the North-Karelia Salt Project in 1979, WHO FINMONICA studies 1982 and 1987, and the National FINRISK Study in 2002. During the follow-up, 161 new cases of diabetes were observed. Compared with the highest quartile of sodium intake (mean 6.7 g/day) persons in the lowest quartile (mean 2.2 g/day) had a lower multivariate adjusted HR of diabetes (HR: 0.52; 95% CI: 0.32-0.87) (92). A systematic review and metaanalyses of 20 randomised studies, 504 healthy hypertensive participants aged 10 to 79 years and with a mean duration of approximately 11 days (range 3 to 56 days) assessed effects of sodium restriction on fasting serum glucose (93). A sodium restriction from a mean of 4.8 to a mean of 1.1 g/day did not change fasting serum glucose. The difference in glucose from low to high sodium intake was 0.01 mmol/l (95% CI: -0.04 to 0.05 mmol/l). Sufficiently powered long-term sodium restriction trials are needed to assess whether low sodium intake can prevent glucose intolerance and type 2 diabetes.

The World Cancer Research Fund/American Institute for Cancer Research (WCRF/AICR) concluded that there is probable evidence that total intake of salt and sodium is associated with stomach cancer (94). A meta-analysis of seven prospective cohort studies found an increased risk with dietary intakes categorised as 'high' or 'moderately high' compared to 'low' intakes (95). The meta-analysis included 10 cohorts, 268,718 participants, and 1,474 events with a follow-up of at least 4 years. The categorisation of intakes was based on reported tertiles or middle and extreme quintiles. Two newer systematic review and meta-analyses replicated the finding of a positive association between salt intake and gastric cancer (96, 97). High salt intake can change the viscosity of mucus, increase gastrin secretion and inflammatory responses, enhance epithelial damage and epithelial cell proliferation, and it may have synergistic effects with Helicobacter pylori infection (98). Damages in the gastric mucosa may make it also vulnerable to carcinogenic compounds in the food.

High salt intake may play a role in the development of autoimmune disease. The studies with autoimmune animal models have shown that a high sodium diet can induce the development of multiple sclerosis (MS), inflammatory bowel disease, rheumatoid arthritis, and lupus nephritis. High salt intake has been suggested to drive lymphocyte differentiation from regulatory T cells towards proinflammatory T-helper (Th)-1 and Th17 phenotypes, and to lead in changes in the gut microbiota (99, 100). Data from human studies of the relations between sodium intake and autoimmune diseases are sparse. One large cross-sectional study from Spain, based on health questionnaires, and several case-control studies suggest that a high salt intake is associated with a higher risk of rheumatoid arthritis (101). The association of salt and MS has been assessed in three studies as the time of remission to relapse of the disease or as the time from the onset of clinically isolated symptoms (CIS) to the onset of multiple signs of the disease (99). One study showed an inverse association between sodium intake and remission time, but the two other studies did not observe any significant associations between sodium intake and remission time or conversion time from CIS to MS.

Requirement and recommended intakes

Adults

Among adults, sodium balance can be maintained at intakes as low as 10 mmol (230 mg) per day, which corresponds to about 0.6 g of salt (102, 103). An intake of 25 mmol (575 mg) per day, corresponding to about 1.5 g salt, is set as the estimated lower intake level and accounts for variations in physical activity and climate (102). This is line with findings from populations using no added salt in their diet.

The average BP is approximately 100/60 mmHg and does not rise with age in populations with a very low sodium intake (<0.8 g/day), and the rise with age is only gently sloping in populations with a sodium intake between 0.8 and 1.3 g/day (44). A positive linear correlation between sodium intake and BP has been confirmed in cross-sectional and observational studies using proper measurement techniques of sodium intake. According to a recent meta-analysis, there is a dose-relationship between sodium reduction and BP decrease and a significant BP decrease from an intake of 2.5 g/day to an intake of 2 g/day (55). Several intervention studies have confirmed the efficiency and safety of reducing blood sodium intake to a level of less than 2 g/day. Analyses assessing sodium intake using at least two 24-h urine samples have shown a linear positive relationship between sodium intake and the risk of a cardiovascular event or death. The optimal sodium intake goal would be probably about 1.5 g/day.

Based on available data, a limitation of sodium intake to 2 g/day – corresponding to 5 g salt (NaCl) – is suggested.

Children

BP rises along with increasing sodium *intake* from early childhood to late adulthood. SBP may already at the age of 15 years exceed by 15–20 mmHg that of adults in populations consuming very low-sodium diets. The BP rise with age is a consequence of structural and functional changes in the vasculature. BP measured in childhood tracks with the level measured in adulthood and predicts early atherosclerosis in adulthood. The available data suggest that a reduction in sodium intake at a young age is associated with lower BP in later life. Following a lifelong salt-reduced

diet beginning in early childhood is recommended. It is also prudent to limit sodium intake in childhood in order to avoid preference for a diet with a high salt content.

Pregnancy and lactation

Pregnancy and lactation are associated with a small increase in the physiological requirements for sodium by about 0.07 g or 3 mmol per day (pregnancy) and 0.13 g or 6 mmol per day (full lactation) (4). These amounts are small and can apparently be handled by the body's homeostatic system. There is a lack of evidence to suggest that sodium requirements during pregnancy and lactation differ significantly from that of non-pregnant women.

International expert reports

The sodium intake recommendation of NNR2023 is in line with the recommendations of WHO (104, 105). EFSA (4) and the National Health and Medical Research Council (NHMRC) of Australia and New Zealand (106). The EFSA Panel considers that 2.0 g sodium/day is safe and adequate intake (AI) for the general European Union population of adults (4). The British Expert Panel have published recommendations to limit salt intake to 6 g/day among adults (107). The expert report of the nutrition societies of Germany, Austria, and Switzerland set a reference value of 1.5 g/day for sodium intake in adults (108). The American consensus study report of the National Academies of Sciences, Engineering and Medicine set also the reference level (AI) of sodium intake of adults to 1.5 g/day, and suggested to reduce the intake if above 2.3 g/ day (19). The American Heart Association (AHA) and the American College of Cardiology (ACC) suggest to reduce sodium intake to less than 1.5 g/day in the prevention and treatment of hypertension (109, 110). The importance of population-wide sodium reduction as a means to prevent CVD and stroke has been pointed out by WHO, the AHA (111), and the British National Institute for Health and Clinical Excellence (112).

Conflict of interest and funding

Member of the Nutrition Council of the Finnish Heart Association, 2000-present.

Member of the Executive Committee of the Finnish Hypertension Society, 1993-present.

Chair of the Hypertension Current Care Guideline working group appointed by the Finnish Medical Society Duodecim and the Finnish Hypertension Society, 2000-present. Funding statement: nothing to declare.

References

Blomhoff R, Andersen R, Arnesen EK, Christensen JJ, Eneroth H, Erkkola M, et al. Nordic Nutrition Recommendations 2023. Copenhagen: Nordic Council of Ministers; 2023. doi: 10.6027/ nord2023-003

- Christensen JJ, Arnesen EK, Andersen R, Eneroth H, Erkkola M, Høyer A, et al. The Nordic Nutrition Recommendations 2022 – principles and methodologies. Food Nutr Res 2020 Jun 18; 64: 4402. doi: 10.29219/fnr.v64.4402
- Høyer A, Christensen JJ, Arnesen EK, Andersen R, Eneroth H, Erkkola M, et al. The Nordic Nutrition Recommendations 2022

 prioritisation of topics for de novo systematic reviews. Food Nutr Res 2021 Oct 8; 65: 7828. doi: 10.29219/fnr.v65.7828
- Turck D, Castenmiller J, de Henauw S, Hirsch-Ernst K-I, Kearney J, Maciuk A, et al. Dietary reference values for sodium. EFSA J 2019; 17(9): 5778 . doi: 10.2903/j.efsa.2019.5778
- Bie P. Mechanisms of sodium balance: total body sodium, surrogate variables, and renal sodium excretion. Am J Physiol Regul Integr Comp Physiol 2018; 315(5): R945–62. doi: 10.1152/ ajpregu.00363.2017
- Kopp C, Linz P, Dahlmann A, Hammon M, Jantsch J, Müller DN, et al. 23Na magnetic resonance imaging-determined tissue sodium in healthy subjects and hypertensive patients. Hypertension 2013; 61(3): 635–40. doi: 10.1161/ HYPERTENSIONAHA.111.00566
- Cogswell ME, Maalouf J, Elliott P, Loria CM, Patel S, Bowman BA. Use of urine biomarkers to assess sodium intake: challenges and opportunities. Annu Rev Nutr 2015; 35: 349–87. doi: 10.1146/annurev-nutr-071714-034322
- Lucko AM, Doktorchik C, Woodward M, Cogswell M, Neal B, Rabi D, et al. Percentage of ingested sodium excreted in 24-hour urine collections: a systematic review and meta-analysis. J Clin Hypertens (Greenwich) 2018; 20(9): 1220–9. doi: 10.1111/ jch.13353
- Freedman LS, Commins JM, Moler JE, Willett W, Tinker LF, Subar AF, et al. Pooled results from 5 validation studies of dietary self-report instruments using recovery biomarkers for potassium and sodium intake. Am J Epidemiol 2015; 181(7): 473–87. doi: 10.1093/aje/kwu325
- McLean RM, Farmer VL, Nettleton A, Cameron CM, Cook NR, Campbell NRC, et al. Assessment of dietary sodium intake using a food frequency questionnaire and 24-hour urinary sodium excretion: a systematic literature review. J Clin Hypertens (Greenwich) 2017; 19(12): 1214–30. doi: 10.1111/ jch.13148
- McLean RM, Farmer VL, Nettleton A, Cameron CM, Cook NR, Woodward M, et al. Twenty-four-hour diet recall and diet records compared with 24-hour urinary excretion to predict an individual's sodium consumption: a systematic review. J Clin Hypertens (Greenwich) 2018; 20(10): 1360–76. doi: 10.1111/ jch.13391
- 12. Campbell NRC, He FJ, Tan M, Cappuccio FP, Neal B, Woodward M, et al. The International Consortium for Quality Research on Dietary Sodium/Salt (TRUE) position statement on the use of 24-hour, spot, and short duration (<24 hours) timed urine collections to assess dietary sodium intake. J Clin Hypertens (Greenwich) 2019; 21(6): 700–9. doi: 10.1111/ jch.13551
- Lerchl K, Rakova N, Dahlmann A, Rauh M, Goller U, Basner M, et al. Agreement between 24-hour salt ingestion and sodium excretion in a controlled environment. Hypertension 2015; 66(4): 850–7. doi: 10.1161/HYPERTENSIONAHA.115.05851
- Rakova N, Jüttner K, Dahlmann A, Schröder A, Linz P, Kopp C, et al. Long-term space flight simulation reveals infradian rhythmicity in human Na(+) balance. Cell Metab 2013; 17(1): 125–31. doi: 10.1016/j.cmet.2012.11.013
- 15. O'Donnell MJ, Yusuf S, Mente A, Gao P, Mann JF, Teo K, et al. Urinary sodium and potassium excretion and risk of

- O'Donnell M, Mente A, Rangarajan S, McQueen MJ, Wang X, Liu L, et al. Urinary sodium and potassium excretion, mortality, and cardiovascular events. N Engl J Med 2014 Aug 14; 371(7): 612–23. doi: 10.1056/NEJMoa1311889. Erratum in: N Engl J Med 2014 Sep 25; 371(13): 1267.
- He FJ, Ma Y, Campbell NRC, MacGregor GA, Cogswell ME, Cook NR. Formulas to estimate dietary sodium intake from spot urine alter sodium-mortality relationship. Hypertension 2019; 74(3): 572–80. doi: 10.1161/HYPERTENSIONAHA.119.13117
- Baxmann AC, Ahmed MS, Marques NC, Menon VB, Pereira AB, Kirsztajn GM, et al. Influence of muscle mass and physical activity on serum and urinary creatinine and serum cystatin C. Clin J Am Soc Nephrol 2008 Mar; 3(2): 348–54. doi: 10.2215/ CJN.02870707
- National Academies of Sciences, Engineering, and Medicine. Dietary reference intakes for sodium and potassium. Washington, DC: The National Academies Press; 2019. doi: 10.17226/25353
- Andersen L, Rasmussen LB, Larsen EH, Jakobsen J. Intake of household salt in a Danish population. Eur J Clin Nutr 2009 May; 63(5): 598–604. doi: 10.1038/ejcn.2008.18
- Amcoff E, Edberg A, Enghardt Barbieri H. Riksmaten vuxna 2010–11. Livsmedels- och näringsintag bland vuxna i Sverige. Resultat från matvaneundersökningen utförd 2010–11 (Food and nutrient intake in Sweden 2010–11). (In Swedish, summary, figures and tables in English). Uppsala: Livsmedelsverket; 2012.
- 22. Valsta L, Kaatinen N, Tapanainen H, Männistö S, Sääksjärvi K. The National FINDIET 2017 Survey. (In Finnish, summary, figures and tables in English). Report No.: 12/2018. Helsinki: National Institute for Health and Welfare (THL); 2018. ISBN 978-952-343-238-3 (online publication).
- Lemming EW, Pitsi T. The Nordic Nutrition Recommendations 2022 – food consumption and nutrient intake in the adult population of the Nordic and Baltic countries. Food Nutr Res 2022 Jun 8; 66: 8572. doi: 10.29219/fnr.v66.8572
- Reinivuo H, Valsta LM, Laatikainen T, Tuomilehto J, Pietinen P. Sodium in the Finnish diet: II trends in dietary sodium intake and comparison between intake and 24-h excretion of sodium. Eur J Clin Nutr 2006 Oct; 60(10): 1160–7. doi: 10.1038/ sj.ejcn.1602431
- 25. Inese S, Lazda I, Goldmanis M. Pētījums par sāls un joda patēriņu Latvijas pieaugušo iedzīvotāju populācijā (Study of salt and iodine consumption in the adult population in Latvia). Riga: Bior; 2020.
- 26. Campbell NRC, Train EJ. A systematic review of fatalities related to acute ingestion of salt. A need for warning labels? Nutrients 2017 Jun 23; 9(7): 648. doi: 10.3390/nu9070648
- 27. Grimes CA, Bolton KA, Booth AB, Khokhar D, Service C, He FH, et al. The association between dietary sodium intake, adiposity and sugar-sweetened beverages in children and adults: a systematic review and meta-analysis. Br J Nutr 2021; 126(3): 409–27. doi: 10.1017/S0007114520004122
- Strazzullo P, Galletti F, Barba G. Altered renal handling of sodium in human hypertension: short review of the evidence. Hypertension 2003; 41(5): 1000–5. doi: 10.1161/01. HYP.0000066844.63035.3A
- Redon J, Cifkova R, Laurent S, Nilsson P, Narkiewicz K, Erdine S, et al. Mechanisms of hypertension in the cardiometabolic syndrome. J Hypertens 2009; 27(3): 441–51. doi: 10.1097/ HJH.0b013e32831e13e5

- Kawarazaki W, Fujita T. The role of aldosterone in obesity-related hypertension. Am J Hypertens 2016; 29(4): 415–23. doi: 10.1093/ajh/hpw003
- Bailey MA. 11β-Hydroxysteroid dehydrogenases and hypertension in the metabolic syndrome. Curr Hypertens Rep 2017; 19(12): 100. doi: 10.1007/s11906-017-0797-z
- 32. GBD 2017 Risk Factor Collaborators. Global, regional, and national comparative risk assessment of 84 behavioural, environmental and occupational, and metabolic risks or clusters of risks for 195 countries and territories, 1990–2017: a systematic analysis for the Global Burden of Disease Study 2017. Lancet 2018; 392: 1923–94. doi: 10.1016/S0140-6736(18)32225-6
- 33. Lewington S, Clarke R, Qizilbash N, Peto R, Collins R, Prospective Studies Collaboration. Age-specific relevance of usual blood pressure to vascular mortality: a meta-analysis of individual data for one million adults in 61 prospective studies. Lancet 2002; 360: 1903–13. doi: 10.1016/S0140-6736(02)11911-8
- 34. Franco OH, Peeters A, Bonneux L, de Laet C. Blood pressure in adulthood and life expectancy with cardiovascular disease in men and women: life course analysis. Hypertension 2005; 46: 280–6. doi: 10.1161/01.HYP.0000173433.67426.9b
- 35. Koponen P, Borodulin K, Lundqvist A, Sääksjärvi K, Koskinen S. The National FinHealth 2017 survey. (In Finnish, summary in English). Report No: 4/2018. Helsinki: National Institute for Health and Welfare (THL); 2018. ISBN 978-952-343-105-8 (online publication)
- 36. NCD Risk Factor Collaboration (NCD-RisC). Worldwide trends in blood pressure from 1975 to 2015: a pooled analysis of 1479 population-based measurement studies with 19·1 million participants. Lancet 2017 Jan 7; 389(10064): 37–55. doi: 10.1016/S0140-6736(16)31919-5. Erratum in: Lancet 2020 Sep 26; 396(10255): 886.
- Grillo A, Salvi L, Coruzzi P, Salvi P, Parati G. Sodium intake and hypertension. Nutrients 2019; 11(9): 1970. doi: 10.3390/ nu11091970
- Marketou ME, Maragkoudakis S, Anastasiou I, Nakou H, Plataki M, Vardas PE, et al. Salt-induced effects on microvascular function: a critical factor in hypertension mediated organ damage. J Clin Hypertens (Greenwich) 2019; 21(6): 749–57. doi: 10.1111/jch.13535
- Yu HC, Burrell LM, Black MJ, Wu LL, Dilley RJ, Cooper ME, et al. Salt induces myocardial and renal fibrosis in normotensive and hypertensive rats. Circulation 1998; 98(23): 2621–8. doi: 10.1161/01.cir.98.23.2621
- Farquhar WB, Edwards DG, Jurkovitz CT, Weintraub WS. Dietary sodium and health: more than just blood pressure. J Am Coll Cardiol 2015; 65(10): 1042–50. doi: 10.1016/j. jacc.2014.12.039
- Kawarazaki W, Fujita T. Role of Rho in salt-sensitive hypertension. Int J Mol Sci 2021 Mar 15; 22(6): 2958. doi: 10.3390/ ijms22062958
- 42. Iwamoto T, Kita S, Zhang J, Blaustein MP, Arai Y, Yoshida S, et al. Salt-sensitive hypertension is triggered by Ca2+ entry via Na+/Ca2+ exchanger type-1 in vascular smooth muscle. Nat Med 2004 Nov; 10(11): 1193–9. doi: 10.1038/nm1118
- 43. Safar ME, Thuilliez C, Richard V, Benetos A. Pressureindependent contribution of sodium to large artery structure and function in hypertension. Cardiovasc Res 2000 May; 46(2): 269–76. doi: 10.1016/s0008-6363(99)00426-5
- 44. Intersalt: an international study of electrolyte excretion and blood pressure. Results for 24 hour urinary sodium and potassium excretion. Intersalt Cooperative Research Group. BMJ 1988 Jul 30; 297(6644): 319–28. doi: 10.1136/bmj.297.6644.319

- 45. Elliott P, Stamler J, Nichols R, Dyer AR, Stamler R, Kesteloot H, et al. Intersalt revisited: further analyses of 24 hour sodium excretion and blood pressure within and across populations. Intersalt Cooperative Research Group. BMJ 1996 May 18; 312(7041): 1249–53. doi: 10.1136/bmj.312.7041.1249
- 46. Law MR, Frost CD, Wald NJ. By how much does dietary salt reduction lower blood pressure? I – analysis of observational data among populations. BMJ 1991 Apr 6; 302(6780): 811–15. doi: 10.1136/bmj.302.6780.811
- 47. Khaw KT, Bingham S, Welch A, Luben R, O'Brien E, Wareham N, et al. Blood pressure and urinary sodium in men and women: the Norfolk Cohort of the European Prospective Investigation into Cancer (EPICNorfolk). Am J Clin Nutr 2004 Nov; 80(5): 1397–403. doi: 10.1093/ajcn/80.5.1397
- Leyvraz M, Chatelan A, da Costa BR, Taffé P, Paradis G, Bovet P, et al. Sodium intake and blood pressure in children and adolescents: a systematic review and meta-analysis of experimental and observational studies. Int J Epidemiol 2018; 47(6): 1796–810. doi: 10.1093/ije/dyy121
- Mente A, O'Donnell MJ, Rangarajan S, McQueen MJ, Poirier P, Wielgosz A, et al. Association of urinary sodium and potassium excretion with blood pressure. N Engl J Med 2014; 371(7): 601–11. doi: 10.1056/NEJMoa1311989
- Newberry SJ, Chung M, Anderson CAM, Chen C, Fu Z, Tang A, et al. Sodium and potassium intake: effects on chronic disease outcomes and risks. Report No.: 18-EHC009-EF. Rockville, MD: Agency for Healthcare Research and Quality (US); 2018.
- 51. Graudal NA, Hubeck-Graudal T, Jurgens G. Effects of low sodium diet versus high sodium diet on blood pressure, renin, aldosterone, catecholamines, cholesterol, and triglyceride. Cochrane Database Syst Rev 2020 Dec 12; 12(12): CD004022. doi: 10.1002/14651858.CD004022.pub5
- Rankin LI, Luft FC, Henry DP, Gibbs PS, Weinberger MH. Sodium intake alters the effects of norepinephrine on blood pressure. Hypertension 1981; 3(6): 650–6. doi: 10.1161/01. hyp.3.6.650
- 53. Skrabal F, Herholz H, Neumayr M, Hamberger L, Ledochowski M, Sporer H, et al. Salt sensitivity in humans is linked to enhanced sympathetic responsiveness and to enhanced proximal tubular reabsorption. Hypertension 1984; 6(2 Pt 1): 152–8. doi: 10.1161/01.HYP.6.2_Pt_1.152
- Aburto NJ, Hanson S, Gutierrez H, Hooper L, Elliott P, Cappuccio FP. Effect of increased potassium intake on cardiovascular risk factors and disease: systematic review and meta-analyses. BMJ 2013; 346: f1378. doi: 10.1136/bmj.f1378
- Filippini T, Malavolti M, Whelton PK, Naska A, Orsini N, Vinceti M. Blood pressure effects of sodium reduction: dose-response meta-analysis of experimental studies. Circulation 2021; 143(16): 1542–67. doi: 10.1161/CIRCULATIONAHA.120.050371
- Jula A, Ronnemaa T, Rastas M, Karvetti RL, Maki J. Longterm nopharmacological treatment for mild to moderate hypertension. J Intern Med 1990 Jun; 227(6): 413–21. doi: 10.1111/j.1365-2796.1990.tb00180.x
- Jula AM, Karanko HM. Effects on left ventricular hypertrophy of long-term nonpharmacological treatment with sodium restriction in mild-to-moderate essential hypertension. Circulation 1994 Mar; 89(3): 1023–31. doi: 10.1161/01.CIR.89.3.1023
- Jula A, Rönnemaa T, Tikkanen I, Karanko H. Responses of atrial natriuretic factor to long-term sodium restriction in mild to moderate hypertension. J Intern Med 1992; 231(5): 521–9. doi: 10.1111/j.1365-2796.1992.tb00968.x
- Appel LJ, Moore TJ, Obarzanek E, Vollmer WM, Svetkey LP, Sacks FM, et al. A clinical trial of the effects of dietary patterns

on blood pressure. DASH Collaborative Research Group. N Engl J Med 1997 Apr 17; 336(16): 1117–24. doi: 10.1056/ NEJM199704173361601

- 60. Sacks FM, Svetkey LP, Vollmer WM, Appel LJ, Bray GA, Harsha D, et al. Effects on blood pressure of reduced dietary sodium and the Dietary Approaches to Stop Hypertension (DASH) diet. DASH-Sodium Collaborative Research Group. N Engl J Med 2001 Jan 4; 344(1): 3–10. doi: 10.1056/ NEJM200101043440101
- Harsha DW, Sacks FM, Obarzanek E, Svetkey LP, Lin PH, Bray GA, et al. Effect of dietary sodium intake on blood lipids: results from the DASH-sodium trial. Hypertension 2004; 43(2): 393–8. doi: 10.1161/01.HYP.0000113046.83819.a2
- 62. Shear CL, Burke GL, Freedman DS, Berenson GS. Value of childhood blood pressure measurements and family history in predicting future blood pressure status: results from 8 years of follow-up in the Bogalusa Heart Study. Pediatrics 1986 Jun; 77(6): 862–9. doi: 10.1542/peds.77.6.862
- Lauer RM, Clarke WR, Mahoney LT, Witt J. Childhood predictors for high adult blood pressure. The Muscatine Study. Pediatr Clin North Am 1993 Feb; 40(1): 23–40. doi: 10.1016/ S0031-3955(16)38478-4
- 64. van den Elzen AP, de Ridder MA, Grobbee DE, Hofman A, Witteman JC, Uiterwaal CS. Families and the natural history of blood pressure. A 27-year follow-up study. Am J Hypertens 2004 Oct; 17(10): 936–40. doi: 10.1016/S0895-7061(04)00871-4
- 65. Niinikoski H, Jula A, Viikari J, Ronnemaa T, Heino P, Lagstrom H, et al. Blood pressure is lower in children and adolescents with a low-saturated-fat diet since infancy: the special Turku coronary risk factor intervention project. Hypertension 2009 Jun; 53(6): 918–24. doi: 10.1161/HYPERTENSIONAHA.109.130146
- 66. He FJ, Marrero NM, MacGregor GA. Salt intake is related to soft drink consumption in children and adolescents: a link to obesity? Hypertension 2008 Mar; 51(3): 629–34. doi: 10.1161/ HYPERTENSIONAHA.107.100990
- 67. Butte NF, Fox MK, Briefel RR, Siega-Riz AM, Dwyer JT, Deming DM, et al. Nutrient intakes of US infants, toddlers, and preschoolers meet or exceed dietary reference intakes. J Am Diet Assoc 2010 Dec; 110(12 Suppl.): S27–37. doi: 10.1016/j. jada.2010.09.004
- Chen X, Wang Y. Tracking of blood pressure from childhood to adulthood: a systematic review and metaregression analysis. Circulation 2008 Jun 24; 117(25): 3171–80. doi: 10.1161/ CIRCULATIONAHA.107.730366
- 69. Juhola J, Magnussen CG, Viikari JS, Kahonen M, Hutri-Kahonen N, Jula A, et al. Tracking of serum lipid levels, blood pressure, and body mass index from childhood to adulthood: the Cardiovascular Risk in Young Finns Study. J Pediatr 2011 Oct; 159(4): 584–90. doi: 10.1016/j.jpeds.2011.03.021
- Berenson GS, Srinivasan SR, Bao W, Newman WP, 3rd, Tracy RE, Wattigney WA. Association between multiple cardiovascular risk factors and atherosclerosis in children and young adults. The Bogalusa Heart Study. N Engl J Med 1998 Jun 4; 338(23): 1650–6. doi: 10.1056/NEJM199806043382302
- Raitakari OT, Juonala M, Kahonen M, Taittonen L, Laitinen T, Maki-Torkko N, et al. Cardiovascular risk factors in childhood and carotid artery intima-media thickness in adulthood: the Cardiovascular Risk in Young Finns Study. JAMA 2003 Nov 5; 290(17): 2277–83. doi: 10.1001/jama.290.17.2277
- Geleijnse JM, Hofman A, Witteman JC, Hazebroek AA, Valkenburg HA, Grobbee DE. Long-term effects of neonatal sodium restriction on blood pressure. Hypertension 1997 Apr; 29(4): 913–17. doi: 10.1161/01.HYP.29.4.913

- He FJ, MacGregor GA. Importance of salt in determining blood pressure in children: meta-analysis of controlled trials. Hypertension 2006 Nov; 48(5): 861–9. doi: 10.1161/01. HYP.0000245672.27270.4a
- 74. Stolarz-Skrzypek K, Kuznetsova T, Thijs L, Tikhonoff V, Seidlerová J, Richart T, et al. Fatal and nonfatal outcomes, incidence of hypertension, and blood pressure changes in relation to urinary sodium excretion. JAMA 2011 May 4; 305(17): 1777–85. doi: 10.1001/jama.2011.574
- Strazzullo P, D'Elia L, Kandala NB, Cappuccio FP. Salt intake, stroke, and cardiovascular disease: meta-analysis of prospective studies. BMJ 2009; 339: b4567. doi: 10.1136/bmj.b4567
- Alderman MH, Madhavan S, Cohen H, Sealey JE, Laragh JH. Low urinary sodium is associated with greater risk of myocardial infarction among treated hypertensive men. Hypertension 1995; 25(6): 1144–52. doi: 10.1161/01.hyp.25.6.1144
- 77. Poggio R, Gutierrez L, Matta MG, Elorriaga N, Irazola V, Rubinstein A. Daily sodium consumption and CVD mortality in the general population: systematic review and meta-analysis of prospective studies. Public Health Nutr 2015; 18(4): 695–704. doi: 10.1017/S1368980014000949
- Wang YJ, Yeh TL, Shih MC, Tu YK, Chien KL. Dietary sodium intake and risk of cardiovascular disease: a systematic review and dose-response meta-analysis. Nutrients 2020; 12(10): 2934. doi: 10.3390/nu12102934
- Ma Y, He FJ, Sun Q, Yuan C, Kieneker LM, Curhan GC, et al. 24-Hour urinary sodium and potassium excretion and cardiovascular risk. N Engl J Med 2022; 386(3): 252–63. doi: 10.1056/ NEJMoa2109794
- Olde Engberink RHG, van den Hoek TC, van Noordenne ND, van den Born BH, Peters-Sengers H, Vogt L. Use of a single baseline versus multiyear 24-hour urine collection for estimation of long-term sodium intake and associated cardiovascular and renal risk. Circulation 2017; 136(10): 917–26. doi: 10.1161/ CIRCULATIONAHA.117.029028
- Neal B, Wu Y, Feng X, Zhang R, Zhang Y, Shi J, et al. Effect of salt substitution on cardiovascular events and death. N Engl J Med 2021; 385(12): 1067–77. doi: 10.1056/NEJMoa2105675
- Pietinen P, Vartiainen E, Seppanen R, Aro A, Puska P. Changes in diet in Finland from 1972 to 1992: impact on coronary heart disease risk. Prev Med 1996 May-Jun; 25(3): 243–50. doi: 10.1006/pmed.1996.0053
- Laatikainen T, Nissinen A, Kastarinen M, Jula A, Tuomilehto J. Blood pressure, sodium intake, and hypertension control: lessons from the North Karelia project. Glob Heart 2016; 11(2): 191–9. doi: 10.1016/j.gheart.2016.04.011
- Laatikainen T, Pietinen P, Valsta L, Sundvall J, Reinivuo H, Tuomilehto J. Sodium in the Finnish diet: 20-year trends in urinary sodium excretion among the adult population. Eur J Clin Nutr 2006 Aug; 60(8): 965–70. doi: 10.1038/ sj.ejcn.1602406
- Vartiainen E, Puska P, Pekkanen J, Tuomilehto J, Jousilahti P. Changes in risk factors explain changes in mortality from ischaemic heart disease in Finland. BMJ 1994 Jul 2; 309(6946): 23–7. doi: 10.1136/bmj.309.6946.23
- Vartiainen E, Sarti C, Tuomilehto J, Kuulasmaa K. Do changes in cardiovascular risk factors explain changes in mortality from stroke in Finland? BMJ 1995 Apr 8; 310(6984): 901–4. doi: 10.1136/bmj.310.6984.901
- Nomura K, Asayama K, Jacobs L, Thijs L, Staessen JA. Renal function in relation to sodium intake: a quantitative review of the literature. Kidney Int 2017; 92(1): 67–78. doi: 10.1016/j. kint.2016.11.032

- D'Elia L, Rossi G, Schiano di Cola M, Savino I, Galletti F, Strazzullo P. Meta-analysis of the effect of dietary sodium restriction with or without concomitant renin-angiotensin-aldosterone system-inhibiting treatment on albuminuria. Clin J Am Soc Nephrol 2015; 10(9): 1542–52. doi: 10.2215/CJN.09110914
- Cappuccio FP, Kalaitzidis R, Duneclift S, Eastwood JB. Unravelling the links between calcium excretion, salt intake, hypertension, kidney stones and bone metabolism. J Nephrol 2000 May–Jun; 13(3): 169–77.
- 90. Afsar B, Kiremit MC, Sag AA, Tarim K, Acar O, Esen T, et al. The role of sodium intake in nephrolithiasis: epidemiology, pathogenesis, and future directions. Eur J Intern Med 2016 Nov; 35: 16–19. doi: 10.1016/j.ejim.2016.07.001. Erratum in: Eur J Intern Med 2017 Jan; 37: 90.
- Tuomilehto J, Jousilahti P, Rastenyte D, Moltchanov V, Tanskanen A, Pietinen P, et al. Urinary sodium excretion and cardiovascular mortality in Finland: a prospective study. Lancet 2001; 357(9259): 848–51. doi: 10.1016/S0140-6736(00)04199-4
- 92. Vuori MA, Harald K, Jula A, Valsta L, Laatikainen T, Salomaa V, et al. 24-h urinary sodium excretion and the risk of adverse outcomes. Ann Med 2020; 52(8): 488–96. doi: 10.1080/07853890.2020.1780469
- Patel SM, Cobb P, Saydah S, Zhang X, de Jesus JM, Cogswell ME. Dietary sodium reduction does not affect circulating glucose concentrations in fasting children or adults: findings from a systematic review and meta-analysis. J Nutr 2015; 145(3): 505– 513. doi: 10.3945/jn.114.195982
- 94. Clinton SK, Giovannucci EL, Hursting SD. The World Cancer Research Fund/American Institute for Cancer Research third expert report on diet, nutrition, physical activity, and cancer: impact and future directions. J Nutr 2020 Apr 1; 150(4): 663–71. doi: 10.1093/jn/nxz268
- 95. D'Elia L, Rossi G, Ippolito R, Cappuccio FP, Strazzullo P. Habitual salt intake and risk of gastric cancer: a meta-analysis of prospective studies. Clin Nutr 2012 Aug; 31(4): 489–98. doi: 10.1016/j.clnu.2012.01.003
- 96. Ge S, Feng X, Shen L, Wei Z, Zhu Q, Sun J. Association between habitual dietary salt intake and risk of gastric cancer: a systematic review of observational studies. Gastroenterol Res Pract 2012; 2012: 808120. doi: 10.1155/2012/808120
- Wu B, Yang D, Yang S, Zhang G. Dietary salt intake and gastric cancer risk: a systematic review and meta-analysis. Front Nutr 2021; 8: 801228. doi: 10.3389/fnut.2021.801228
- Vahid F, Davoodi SH. Nutritional factors involved in the etiology of gastric cancer: a systematic review. Nutr Cancer 2021; 73(3): 376–90. doi: 10.1080/01635581.2020.1756353
- 99. Sharif K, Amital H, Shoenfeld Y. The role of dietary sodium in autoimmune diseases: the salty truth [published correction appears in Autoimmun Rev 2019 Feb; 18(2): 214]. Autoimmun Rev 2018; 17(11): 1069–73. doi: 10.1016/j.autrev.2018.05.007
- 100. Haase S, Wilck N, Kleinewietfeld M, Müller DN, Linker RA. Sodium chloride triggers Th17 mediated autoimmunity. J Neuroimmunol 2019; 329: 9–13. doi: 10.1016/j.jneuroim.2018.06.016
- 101.Salgado E, Bes-Rastrollo M, de Irala J, Carmona L, Gómez-Reino JJ. High sodium intake is associated with self-reported rheumatoid arthritis: a cross sectional and case control analysis within the SUN cohort. Medicine (Baltimore) 2015; 94(37): e0924. doi: 10.1097/MD.0000000000924

- 102. European Communities Commission. Reports of the Scientific Committee for Food (Thirty-first series). Nutrient and energy intakes for the European Community. Food – Science and technique series. Luxembourg: Office for Official Publications of the European Communities; 1993.
- 103. Dahl LK. Possible role of salt intake in the development of essential hypertension. 1960. Int J Epidemiol 2005; 34(5): 967– 78. doi: 10.1093/ije/dyh317
- 104.James WPT, Ferro-Luzzi A, Isaksson B, Szostak WB. Healthy nutrition. Report No.: 24. Copenhagen: WHO; 1988.
- 105. WHO. Guideline: sodium intake for adults and children. Geneva: World Health Organization (WHO); 2012.
- 106. National Health and Medical Research Council, Australian Government Department of Health and Ageing, New Zealand Ministry of Health. Nutrient reference values for Australia and New Zealand. Canberra: National Health and Medical Research Council; 2006. Version 1.2. Updated September 2017.
- 107. The Scientific Advisory Committee on Nutrition. Salt and health. London: The Stationery Office; 2003.
- 108.Strohm D, Bechthold A, Ellinger S, Leschik-Bonnet E, Stehle P, Heseker H, et al. Revised reference values for the intake of sodium and chloride. Ann Nutr Metab 2018; 72(1): 12–17. doi: 10.1159/000484355
- 109. Lloyd-Jones DM, Hong Y, Labarthe D, Mozaffarian D, Appel LJ, Van Horn L, et al. Defining and setting national goals for cardiovascular health promotion and disease reduction: the American Heart Association's strategic Impact Goal through 2020 and beyond. Circulation 2010; 121(4): 586–613. doi: 10.1161/CIRCULATIONAHA.109.192703
- 110. Arnett DK, Blumenthal RS, Albert MA, Buroker AB, Goldberger ZD, Hahn EJ, et al. 2019 ACC/AHA guideline on the primary prevention of cardiovascular disease: a report of the American College of Cardiology/American Heart Association Task Force on Clinical Practice Guidelines [published correction appears in Circulation 2019 Sep 10; 140(11): e649–50] [published correction appears in Circulation 2020 Jan 28; 141(4): e60] [published correction appears in Circulation 2020 Apr 21; 141(16): e774]. Circulation 2019; 140(11): e596–46. doi: 10.1161/ CIR.000000000000678
- 111. Appel LJ, Frohlich ED, Hall JE, Pearson TA, Sacco RL, Seals DR, et al. The importance of population-wide sodium reduction as a means to prevent cardiovascular disease and stroke: a call to action from the American Heart Association. Circulation 2011 Mar 15; 123(10): 1138–43. doi: 10.1161/CIR.0b013e31820d0793
- 112.NICE National Institute for Health and Care Excellence. Cardiovascular disease prevention. NICE Guideline PH25. Manchester: National Institute for Health and Clinical Excellence; 2010.

*Antti Jula

Department of Clinical Medicine University of Turku Turku Finland Email: antjul@utu.fi